\documentclass[12pt]{article}

\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amsthm}
\usepackage{amstext}
\usepackage{amssymb}

\newcommand{\N}{\mathbb{N}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\R}{\mathbb{R}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\HH}{\mathbb{H}}


% define some short cuts for other things
% arrows
\newcommand{\ra}{\rightarrow}
\newcommand{\Ra}{\Rightarrow}
\newcommand{\lra}{\longrightarrow}
\newcommand{\Lra}{\Longrightarrow}
\newcommand{\llra}{\longleftrightarrow}
\newcommand{\la}{\leftarrow}
\newcommand{\La}{\Leftarrow}
\newcommand{\lla}{\longleftarrow}
\newcommand{\Lla}{\Longleftarrow}
\newcommand{\da}{\downarrow}
\newcommand{\Da}{\Downarrow}
\newcommand{\ua}{\uparrow}
\newcommand{\Ua}{\Uparrow}
\newcommand{\actson}{\curvearrowright}
\newcommand{\thra}{\twoheadrightarrow}

% mbox's
\newcommand{\aut}{\mbox{Aut}}
\newcommand{\ch}{\mbox{Ch}}
\newcommand{\const}{\mbox{const}}
\newcommand{\diag}{\mbox{diag}}
\newcommand{\dist}{\mbox{dist}}
\newcommand{\id}{\mbox{id}}
\newcommand{\isom}{\mbox{Isom}}
\newcommand{\tr}{\mbox{tr}}

% others
\newcommand{\sm}{\setminus}
\newcommand{\lie}{\mathfrak}


% theorems and such
\newtheorem{prob}{Problem}
\newtheorem{defn}{Definition}[section]
\newtheorem{lemma}[defn]{Lemma}
\newtheorem{property}[defn]{Property}
\newtheorem{thm}[defn]{Theorem}
\newtheorem{cor}[defn]{Corollary}
\newtheorem{rem}[defn]{Remark}
\newtheorem{prop}[defn]{Proposition}
\newtheorem{ques}[defn]{Question}

\begin{document}

\title{Log review}
\date{}
\author{}
\maketitle

\begin{enumerate}
\item Be familiar with exponential equations and how to use them.
Examples: section~5.1(35,45,47)

\item  Be familiar with logarithms.  
Be able to convert exponential equations to logarithm equations,
and logarithm equations to exponential equations:
section~5.2(1-12).

\item  Be familiar with properties of logarithms (pp~370-372).
These are the only manipulations of logarithms you are allowed to do.
Any time you manipulate a logarithm, you must be able to say which rule you used.
Section~5.2(19-36).

\item  Be able to understand logarithms in story problems.
Section~5.2(48,49,56-59).

\item  Be able to solve logarithm and exponential equations.
Section~5.2(all).
Understand even the complicated ones where you have to do more then just 
take the log of both side.  For example section~5.2(example~4,5), 
and exercises~(9-16).

\end{enumerate}







%\bibliographystyle{plain}
%\bibliographystyle{alpha}
%\bibliography{mybib}

\end{document}